RESEARCH ARTICLE

Inference is bliss: Simulation for power estimation for an observational study of a cholera outbreak intervention

Ruwan Ratnayake^{1,2,3}*, Francesco Checchi¹, Christopher I. Jarvis^{1,2}, W. John Edmunds^{1,2}, Flavio Finger³

 Department of Infectious Disease Epidemiology, Faculty of Epidemiology and Population Health, London School of Hygiene and Tropical Medicine, London, United Kingdom, 2 Centre for the Mathematical Modelling of Infectious Diseases, London School of Hygiene and Tropical Medicine, London, United Kingdom, 3 Epicentre, Paris, France

* ruwan.ratnayake@lshtm.ac.uk

Abstract

Background

The evaluation of ring vaccination and other outbreak-containment interventions during severe and rapidly-evolving epidemics presents a challenge for the choice of a feasible study design, and subsequently, for the estimation of statistical power. To support a future evaluation of a *case-area targeted intervention* against cholera, we have proposed a prospective observational study design to estimate the association between the strength of implementation of this intervention across several small outbreaks (occurring within geographically delineated clusters around primary and secondary cases named 'rings') and its effectiveness (defined as a reduction in cholera incidence). We describe here a strategy combining mathematical modelling and simulation to estimate power for a prospective observational study.

Methodology and principal findings

The strategy combines stochastic modelling of transmission and the direct and indirect effects of the intervention in a set of rings, with a simulation of the study analysis on the model results. We found that targeting 80 to 100 rings was required to achieve power \geq 80%, using a basic reproduction number of 2.0 and a dispersion coefficient of 1.0–1.5.

Conclusions

This power estimation strategy is feasible to implement for observational study designs which aim to evaluate outbreak containment for other pathogens in geographically or socially defined rings.



G OPEN ACCESS

Citation: Ratnayake R, Checchi F, Jarvis CI, Edmunds WJ, Finger F (2022) Inference is bliss: Simulation for power estimation for an observational study of a cholera outbreak intervention. PLoS Negl Trop Dis 16(2): e0010163. https://doi.org/10.1371/journal.pntd.0010163

Editor: Ruifu Yang, Beijing Institute of Microbiology and Epidemiology, CHINA

Received: August 16, 2021

Accepted: January 11, 2022

Published: February 16, 2022

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0010163

Copyright: © 2022 Ratnayake et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: No data was used for this study. The R code for the simulations is available here: https://github.com/ruwanepi/CATIpower-sim-shared.git. **Funding:** RR is funded by a Doctoral Foreign Study Award from the Canadian Institutes of Health Research (award number DFS-164266, https:// cihr-irsc.gc.ca/). FC CIJ, and WJE are funded by UK Research and Innovation as part of the Global Challenges Research Fund (grant number ES/ P010873/1, https://www.ukri.org/). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

From Ebola virus disease outbreaks to the COVID-19 pandemic, the use of real-time evaluations of interventions to contain outbreaks is vital for rapidly estimating impact during the outbreak itself. Such evaluations must be both epidemiologically rigorous and logistically feasible to justify their conduct during an outbreak. In this short report, we report on the process (with R code) and the results of a simulation strategy that we devised for power estimation for a prospective observational study of a novel intervention ("case-area targeted intervention") to contain cholera case clusters that present at the start of a new outbreak. We used simulation in two ways: mathematical modelling to simulate the impacts of a cholera outbreak and the intervention, and simulation of the study analysis on the model results. The strategy provided estimates of the sample sizes of study units required to achieve 80% and 90% power. Our findings reinforce that this process is feasible to implement for similar observational study designs which aim to evaluate outbreak containment for other pathogens in geographically or socially defined rings.

Introduction

Fast and efficient disease control approaches are critical for controlling cholera epidemics. Case area-targeted interventions (CATI) aim to interrupt transmission within small cholera outbreaks by rapidly addressing different routes of infection with multiple interventions (i.e., antibiotic chemoprophylaxis, household water treatment, and oral vaccination) in geographical 'rings' of 100–250 metres around the household of the index case. [1,2] Such containment strategies for small outbreaks target people at the highest risk of infection and may be less resource-intensive and more effective than mass, community-wide campaigns over large geographical areas. [1]

We designed an observational study to measure the effects of CATI during a future cholera epidemic response, to be conducted by Médecins Sans Frontières. The evaluation of CATI presents several challenges for the choice of a feasible study design and subsequently, for the estimation of statistical power. Randomizing individuals or communities to different interventions or a placebo is often not feasible and ethically problematic during a demanding epidemic response in a low-resource setting. For the evaluation of ring vaccination with a new vaccine during the 2016 Ebola epidemic in Guinea, an adapted cluster randomized-controlled trial (RCT) design was developed wherein each ring of contacts of confirmed cases was randomized to a different delay to implementation, thereby producing intervention and control groups. [3] During a cholera outbreak, where a package of routine rather than novel interventions is applied, the objective is to assess the allocation strategy. For this question, an RCT design may not always be appropriate or feasible.

Here, a prospective observational study design is considered, where participants or groups are not randomized and the outcome is measured prospectively. [4] In our example, the measurement of effectiveness (i.e., incidence) is related to the strength of implementation of the intervention across small outbreaks rather than an assigned presence or absence of the exposure (i.e., CATI). The strength of implementation is represented by the natural delay between case notification and the implementation of CATI, which may differ across several small outbreaks. This results in CATI rings categorized by the delay between case notification and implementation. Two interrelated challenges emerge, which do not fit well with a classical statistical approach for study design. First, the analysis does not conform to the conventional formulae for sample size and power estimation given the presence of several 'natural' control

groups. Second, the non-independence of infection risk between persons drives the incidence and is difficult to estimate *a priori*. [5] The interventions produce direct and indirect effects on infection and transmission, with infection prevention, infection, and/or treatment of one person affecting the outcome of another person. [6] Moreover, the cumulative effects of a package of interventions are difficult to predict.

In this report, we describe a strategy to estimate power for a prospective observational study across a range of sample sizes. This approach combines stochastic simulation modelling of small outbreaks and the direct and indirect effects of the intervention, with a simulation study of the study analysis based on model results. While simulation studies are often conducted to estimate power for RCTs, there is little documentation of (a) simulation used for other study designs and (b) mathematical modelling to simulate transmission dynamics for power estimation. [7–9] We provide details of the approach, and R code, as a foundation for further application to outbreak intervention studies of other pathogens.

Methods

We describe the study design for which we are calculating power. We then describe the simulation study using the *Aim*, *Data Generating Mechanism*, *Estimand*, *Methods*, *Performance Measures* (ADEMP) framework for the coherent reporting of simulation studies. [7]

Summary of the prospective observational study design

The impact of CATI (which includes single-dose oral cholera vaccination (OCV), point-of-use water treatment, and antibiotic chemoprophylaxis) will be measured by the reduction in the incidence of cholera around the index cases of small outbreaks through direct and indirect protection, as a function of the time to implementation of CATI (Fig 1). The intervention is triggered when a suspected case is detected and tests positive by an enriched rapid diagnostic test (RDT). [10] Then, a 100–250 metre radius around the index case's household is outlined



https://doi.org/10.1371/journal.pntd.0010163.g001

(hereafter, the 'ring'), wherein CATI is rapidly implemented. While the first outbreak clusters may be responded to very rapidly, as the size of the epidemic increases logistical barriers for field teams are anticipated to result in delays to implementation in new rings of up to 7 days, thus creating natural control groups. The ring is the unit of analysis. A regression analysis will model the observed incidence of enriched RDT-positive cholera in rings relative to the time to response (in days) and coverage. The regression function quantifying the association between timeliness/coverage and incidence provides a measure of effectiveness at different levels of performance.

Summary and rationale for the simulation

The study design rests on the assumption that the regression analysis will have sufficient power to detect an association between CATI performance and the incidence of cholera, i.e. that data from a sufficient number of rings of given size and characteristics (e.g. transmissibility of cholera within the rings) will be available. The **aim** of the simulation study is to explore these sample size requirements. We chose as a **data-generating mechanism** a stochastic transmission model to predict the incidence and the direct and indirect effects of CATI, applied with varied delays, on transmission across a large set of rings during a 30-day period. This mechanistic model of transmission and the predicted effect of the intervention is driven by transmission dynamics and is therefore more realistic than the assignment of an effect size, as typically used in statistical simulations. [5] The basic reproduction number for cholera, R_0 , is varied in the modelling scenarios. The estimand is the incidence of suspected cholera in each ring in the first 30 days after presentation of the index case. The **method** used for simulation involves a regression applied to each set of simulated data to estimate the association of CATI performance and cholera incidence, while tracking *p*-values for the association. For each combination of simulation parameters, the mechanistic model and regression on the resulting simulated data are replicated by an assigned number of simulations (n_{sim}) by randomly sampling without replacement over the anticipated number of rings in the study ($n_{\rm rings}$). The proportion of runs in which the regression yields a significant association provides a measure of power given that a sample of n_{rings} are available. A range of n_{sim} (1000–3000) is used to assess the stability of power estimates. The performance measure is the predicted power.

All analyses were carried out in R version 4.0.5, using the following packages: bpmodels [11] for branching process modelling, lme4 for generalized linear mixed modelling, and the map_dfr() function of the purrr package to repetitively apply functions for the simulation coding.[12] In the following sections, we describe each step in detail, which together with the code provided, can be used to replicate the simulation (https://github.com/ruwanepi/CATI-powersim-shared.git).

Stochastic transmission model

Using the bpmodels package, we applied a branching process model which generated infected persons and accounted for the depletion of susceptible persons to produce the incidence for each of 100,000 rings in the first 30 days after notification of the index case. The population size of the ring (normal distribution with mean 500, SD 50) was within the range of the number of people living in a 100–250 metre radius in major African cities including N'Djamena, Conakry, and Lumumbashi (mean 295, range 55–456 persons). [2] People were assumed to mix homogeneously. Given the efficiency of person-to-person and environmentally-mediated cholera transmission within households, there is potential for exponential growth, mediated by the depletion of susceptibles, before effective control measures are implemented. [13] We assumed that the first notified index case was the true primary case for the outbreak and that

all infectious cases were symptomatic and detectable, with some delay. Infection-to-reporting delays before and after CATI implementation were set as Poisson-distributed. The main outcomes of the model were, by ring, (a) cumulative incidence at day 30, and (b) a random effect accounting for the varying delay from infection to reporting of the primary case in the model (categorized as 0, 1, \geq 1 days), as a proxy of the surveillance capacity by geographic area.

To model transmission, the parameters listed in Table 1 were used, and were either sampled from the underlying distributions or fixed. All persons were assumed to be susceptible without immunity derived from previous vaccination or exposure to V. cholerae. An outbreak started with a single seed case and each case generated a number of secondary cases drawn from a negative binomial distribution $Z \sim NegB(R_0, k)$. The mean is equal to the basic reproduction number at the early phase of the outbreak among an unvaccinated population ($R_0 = 1.5, R_0 =$ 2.0). [14,15] Heterogeneity in the number of new infections produced by each individual is represented by a dispersion coefficient (D = 1.0, D = 1.5), which relates to the dispersion parameter of the negative binomial distribution, $k \left(k = \frac{R_0}{D-1}\right)$. [16,17] Each potential new infection was assigned a time of infection drawn from the serial interval distribution, S ~ gamma (shape = 0.5, rate = 0.1). [14] The number of susceptible persons in the population was progressively reduced due to infection or immunity, reducing the mean of the negative binomial offspring distribution by a factor n/N, where n is the number of remaining susceptible and N is the total population, while keeping the dispersion coefficient constant, and truncating the distribution at n. We assumed that no other interventions were implemented before CATI. Four scenarios using high and low R_0 and D were modeled (Table 1).

CATI interventions were then simulated, with a delay from notification of the index case as determined by a Poisson distribution, and the upper limit approximately based on the 75th percentile of the median delay from symptom onset to case presentation derived from a meta-

 Table 1. Parameters for the stochastic transmission model.

Parameter	Values	Reference		
Sampled	Mean (SD)			
Serial interval, days	5 (8), by negative binomial distribution	Azman et al, 2016[14]		
Reporting delay (before CATI), days	1 (0.9), by Poisson distribution ($\lambda = 1$)	Assumed		
Reporting delay (after CATI), days	0.5 (0.7), by Poisson distribution ($\lambda = 0.5$)	Assumed		
Implementation delay, days	3 (1.9), by Poisson distribution ($\lambda = 1.4$)	Ratnayake et al, 2020 [18]		
Population size of ring ± SD	500 (50), by normal distribution	Finger et al, 2019[2]		
Fixed	Values			
Basic reproduction number for index cases, R_0	1.5, 2.0	Azman et al, 2016[14] Camacho et al, 2018[15]		
Dispersion coefficient, D	1.0, 1.5	Emch et al, 2008[16]		
Initial immune, persons, %	0%	Assumed		
Implementation duration, days	1 (main analysis), 2	Ouamba et al, 2021[20]		
Population coverage, %	80% (main analysis), 50%, 60%, 70%	Parker et al, 2017[19]		
Efficacy of antibiotics, %	66%	Reveiz et al, 2001[23]		
Efficacy of water treatment, %	26%	Fewtrell et al, 2005[21]		
Efficacy of safe water storage, %	21%	Roberts et al, 2001[22]		
Efficacy of vaccination, %	87%	Azman et al, 2016[24]		

During each simulation, sampled values are probabilistically sampled and fixed values remain constant. Median (SD), single values, or proportion efficacy are given. Efficacy measures are summarized in Ratnayake et al, 2021.[1]

https://doi.org/10.1371/journal.pntd.0010163.t001

analysis of cholera outbreaks (0 to 5 days), assuming that the surveillance set-up for CATI will prevent longer delays. [18] We assumed that implementation took one day and the population-based coverage was 80%. [19,20] CATI included distribution of (1) water, sanitation, and hygiene (WASH) materials including chlorine tablets and a narrow-neck container so that the efficacy in reducing bacterial concentration via household water treatment (26%) and safe storage (21%) remained consistent for the 30-day period (cumulative efficacy, 41.5%). [21,22]; single-dose, oral antibiotic chemoprophylaxis against infection so that the efficacy in preventing infection (66%) was maintained for the first 2 days, whereafter it loses effect due to its biological half-life [2,23]; and single-dose, oral cholera vaccination (OCV) prevented infection with an efficacy of 87% over a 2-month period, taking effect 7-11 days after administration when peak vibriocidal response is reached. [24] The effectiveness (efficacy*coverage) was calculated in three phases over the 30 days reflecting the plausible timespan over which the relative effects of each intervention would manifest: (1) days 1 to 2: WASH and antibiotic chemoprophylaxis, (2) days 3 to 6: WASH only, (3) days 7 to 30: WASH and vaccination. The combined effect of concurrent interventions was computed as (1 - ((1-effect.A)*(1-effect.B)*...*(1-effect. Z))).

We conducted two sensitivity analyses to explore the main assumptions of rapid 1-day duration of implementation and population coverage of 80%. We evaluated a longer duration of implementation of 2 days and lower population-based coverage estimates of 50%, 60%, and 75%, based on findings from field studies. [19,20].

Simulation method

A generalized linear mixed model (GLMM) was used to estimate the response variable (cumulative incidence rate) as a function of time to implementation. A negative binomial distribution accounted for overdispersion. Fixed effects quantified the main explanatory variable: the overall effect of time from case presentation to CATI implementation. The logarithm of the ring population was used as an offset to produce an incidence rate ratio (IRR). A random effect accounted for the delay from infection to presentation of the index case, which was categorized into 3 classes (0, 1, \geq 1 days). For simplicity, other potential confounders that would require explicit measurement of geographical locations of rings were not considered (e.g., distance between the ring and the base of the field team). Model fit was assessed using the ratio of sum of squares of Pearson residuals to the residual degrees of freedom (to estimate overdispersion), inspection of the width of confidence intervals, and plotting of response by random effect levels (to estimate the benefit of including the random effect, as compared to using a generalized linear model (GLM)).

As the health of individuals in the same ring may be correlated, regression modeling approaches that account for the clustered nature of the data should be used for the study analysis. This includes GLMM, generalized estimating equations (GEE) and generalized additive models (GAM). GLMM uses random effects to account for contextual factors from the rings that alter the relationship between the exposure and the population effect, while GEEs infer the population-averaged effect across all rings. [25] As we expect there will be variance between rings and we may want to explore it further, GLMMs are preferred over GEEs for this study. GAMs add together the non-parametric and parametric fits of separate regressors into a transformed regression. [26] In this study, GAMs may be used if the observed relationship between delay to response and incidence offers a better fit than a purely parametric GLMM model. Regardless of model choice, the effect estimates should remain similar and unbiased.

The expected power was estimated for a range of sample sizes ($n_{rings} = 50-150$ rings), based on recent CATI experiences during large epidemics in Nepal and Haiti, with a target of 80%

Table 2. Parameters for the simulation study.

Parameter	Value	Reference
Number of rings produced by stochastic model	100,000	Assumed
Number of rings randomly sampled (n_{rings})	50, 75, 100, 125, 150	Roskosky et al, 2019[27] Michel et al, 2019[28]
Number of simulations run for each value of n_{obs} (n_{sim})	500, 1,000, 3,000	Morris et al, 2019[7]

https://doi.org/10.1371/journal.pntd.0010163.t002

power. [27,28] We simulated 100,000 CATI rings using the above-described method. We then conducted a simulation study by randomly sampling a set number of rings (n_{rings}), a set number of times (n_{sim}). A negative binomial GLMM was run on each set of n_{rings} . Power was assessed as the number of simulations with a significative effect of delay to CATI implementation (p<0.05), considered to be true positives, divided by the number of n_{sim} , n_{rings} was varied to assess the effect of the number of rings in each study on power. Table 2 lists the simulation parameters including the range of n_{sim} values used to demonstrate consistency in results. For each set of n_{rings} rings randomly sampled without replacement from the 100,000 rings simulated by the stochastic model, 500–3,000 simulations were run to evaluate the reliability of the results.

Findings

Using $R_0 = 2.0$, D = 1.5 and $n_{sim} = 100,000$, the mean caseload increased with each single day from 12 cases (with delays of 0 days) to 59 cases (with delays of 7 days). A higher proportion of outbreaks were extinct by day 30 for the \leq 3-day category versus the >3-day category. An IRR of 1.27 (95% CI 1.25–1.29) was produced, demonstrating a 27% increase in the incidence rate per single day increase in the delay of implementation of CATI (visualized in Fig 2). The model fit is described in S1 Text.



Fig 2. Boxplots of the attack rate of cholera cases (per 1000 population, on a log10 scale) categorized by the delay to CATI implementation (in days) using 100,000 rings (with generalized linear model of the association outlined in orange).

https://doi.org/10.1371/journal.pntd.0010163.g002

				Number of rings					
R _o	D	Duration	Coverage	50	75	80	100	125	150
2	1.5	1	80%	52.4	71.7	73.7	80.6	88.7	94.7
2	1	1	80%	57.3	77.1	81.2	85.8	92.7	96.2
1.5	1.5	1	80%	33.6	37.4	49.5	49.7	58.3	62.8
2	1.5	2	80%	44.4	60.4	60.5	69.5	78.7	84.7
2	1.5	1	50%	52.9	64.2	68.9	77.5	85.9	92.4
2	1.5	1	60%	51.3	64.6	70.6	76.8	85.3	92.3
2	1.5	1	75%	53.6	68.0	72.8	79.1	84.7	92.1

Table 3. Power estimates from main simulations and sensitivity analyses. Shading indicates the variable that was changed (grey), and where power estimates were farthest from the 80% target (\leq 69%, in orange), close to the target (\geq 70 to 79%, in light green), and at or above the target (\geq 80%, in dark green).

https://doi.org/10.1371/journal.pntd.0010163.t003

The compiled power estimates are presented in Table 3 and the main power estimations where R_0 , D, and n_{rings} were varied are displayed in Fig 3 (additional graphs are found in Figs A—D in S1 Text). Using the main model ($R_0 = 2.0$ and D = 1.5), the simulation returned 80.6% (95% CI 71.2–87.6) power with $n_{rings} = 100$; 73.7% power with $n_{rings} = 80$; and, 88.7% power with $n_{rings} = 125$ (Fig 3). Using $R_0 = 2.0$ and D = 1.0, the simulation reached 81.2% (95%



CI 71.9–88.1) power with $n_{rings} = 80$. With $R_0 = 1.5$ and D = 1.5, the power was reduced substantially wherein $n_{rings} = 150$ produced 62.8% power. The model for $R_0 = 1.5$ and D = 1.0 did not converge for any number of rings tested and is omitted from Fig.3. The results were generally consistent when n_{sim} was varied.

Sensitivity analyses

Applying a slower duration of implementation of 2 days meant that 80% power is reached with >125 rings. Using lower population coverage of 50% and 60% increased the sample size required to >100 rings to reach 80% power. Lowering slightly the population coverage to 75% resulted in 79.1% power reached with 100 rings, which is close to the target of 80% power.

Discussion

Our simulation strategy provided a relatively simple means of estimating power and associated sample sizes for an observational study of CATI. Based on an R_0 of 2.0, the sample size required to reach 80% power was 80–100 rings, which was generally maintained when population coverage decreased from 80% to 75%. This would have been feasible during recent experiences in implementing CATI during large epidemics in Kathmandu Valley, Nepal (169 rings in 7 months), and Centre Department, Haiti (238 rings in 24 months). [27,28] Alternately, where CATI was used to suppress the tails of large outbreaks though only at the end of mass vaccination campaigns in Juba, South Sudan and Kribi, Cameroon, the sample size would far exceed the number of rings that are typically implemented. [19,20] As cholera epidemics frequently remain small due to the burn-out of the susceptible pool [18], overdispersion of R_0 leading to extinction [16,17], or the impact of the interventions themselves, a pooled analysis of multiple epidemics within a country implementing the same CATI package could be a more secure prospect to attain the required sample size.

A strength of this simulation strategy is the inclusion of a realistic depiction of CATI implementation which models the relative effects of its composite interventions over time. This accounted for the time-limited effects of antibiotics (~ 2 days) and the 7–11 day delay to a measurable immune response after administration of vaccination. [2,24] Another strength was that the stochastic model accounted for the depletion of susceptible persons to provide a plausible representation of early epidemic growth in a small population. This approach can be adjusted using the real-time estimates of the effective reproduction number (R_E) to update sample size estimation. It is also computationally-light, as the process takes less than 2 hours to run without the use of parallel computing.

There are key limitations to our methodology and its simplifying assumptions. The mean population size of 500 persons reflects urban settings. However, cholera epidemics can occur across urban and rural areas simultaneously and would include smaller rings with lower intracluster variation in incidence. As such, a larger sample size may be required to reach 80% power. For the stochastic model, several parameters relating to the early growth of a cholera epidemic are uncertain. The main model used a relatively high R_0 (2.0) sourced from early epidemic growth in unvaccinated populations in South Sudan and Yemen; considerably lower power was achieved with $R_0 = 1.5$. [14,15] In addition, we assumed the entire population was susceptible at the start of the outbreak, which may not be the case in cholera-endemic or previously-vaccinated areas, lowering the R_E and the measurable effect of CATI. The stochastic model is not spatially-explicit, so transmission between communities is not accounted for, nor is the force of infection external to a given ring which could represent long-distance transmission from outside the community or contamination of the local water supply. [2] A duration of implementation of a single day has been shown in Cameroon and South Sudan [19,20], but

this may not be sufficient to cover the entire ring. This potentially leads to an overestimation of the effect, with the sensitivity analysis finding higher sample size requirements. Outbreak simulations are right censored at 30 days, and thus we cannot determine from the 30 day analysis alone whether outbreaks are fully extinct. How the delay to case detection was parameterized as a random effect may not truly represent the surveillance capacity, indicating that it must be accounted for empirically in the actual analysis of the study. Similarly, key co-variates that are thought to be influential on ring incidence (i.e., coverage, average rainfall, distance from roads) could not be simulated realistically without a more complex, spatially-explicit transmission model.

Despite its limitations, the strategy demonstrates a relatively simple and efficient approach to integrating dynamic modeling of a cholera outbreak with study simulation to guide the design of a prospective observational study that we intend to implement. The approach can be used to provide power estimates for evaluations of similar highly targeted interventions for epidemic-prone diseases delivered rapidly to high-risk communities during an outbreak.

Supporting information

S1 Text. Model fit results and sensitivity analyses. Table A. Power estimates from main simulations and sensitivity analyses Fig A: power estimates using a duration of implementation of two days Fig B: power estimates using a population coverage of 50% Fig C: power estimates using a population coverage of 60% Fig D: power estimates using a population coverage of 75%.

(DOCX)

Acknowledgments

We thank Sebastian Funk (LSHTM) for his advice on the bpmodels package.

Author Contributions

Conceptualization: Ruwan Ratnayake, Francesco Checchi, W. John Edmunds, Flavio Finger.

Data curation: Ruwan Ratnayake.

Formal analysis: Ruwan Ratnayake, Flavio Finger.

Funding acquisition: Ruwan Ratnayake.

Investigation: Ruwan Ratnayake, Flavio Finger.

Methodology: Ruwan Ratnayake, Francesco Checchi, Christopher I. Jarvis, W. John Edmunds, Flavio Finger.

Project administration: Ruwan Ratnayake.

Resources: Ruwan Ratnayake.

Software: Ruwan Ratnayake, Flavio Finger.

Supervision: Francesco Checchi, W. John Edmunds, Flavio Finger.

Validation: Ruwan Ratnayake, Flavio Finger.

Visualization: Ruwan Ratnayake.

Writing - original draft: Ruwan Ratnayake, Flavio Finger.

Writing – review & editing: Francesco Checchi, Christopher I. Jarvis, W. John Edmunds, Flavio Finger.

References

- Ratnayake R, Finger F, Azman AS, Lantagne D, Funk S, Edmunds WJ, et al. Highly targeted spatiotemporal interventions against cholera epidemics, 2000–19: a scoping review. Lancet Infect Dis. 2021; 21 (3):e37–e48. https://doi.org/10.1016/S1473-3099(20)30479-5 PMID: 33096017.
- Finger F, Bertuzzo E, Luquero FJ, Naibei N, Toure B, Allan M, et al. The Potential impact of case-area targeted interventions in response to cholera outbreaks: A modeling study. PLoS Med. 2018; 15(2): e1002509. https://doi.org/10.1371/journal.pmed.1002509 PMID: 29485987; PubMed Central PMCID: PMC5828347.
- 3. Ebola ca Suffit Ring Vaccination Trial Consortium. The Ring vaccination trial: a novel cluster randomised controlled trial design to evaluate vaccine efficacy and effectiveness during outbreaks, with special reference to Ebola. BMJ. 2015; 351:h3740. https://doi.org/10.1136/bmj.h3740 PMID: 26215666; PubMed Central PMCID: PMC4516343.
- Berger ML, Dreyer N, Anderson F, Towse A, Sedrakyan A, Normand SL. Prospective observational studies to assess comparative effectiveness: the ISPOR good research practices task force report. Value Health. 2012; 15(2):217–30. https://doi.org/10.1016/j.jval.2011.12.010 PMID: 22433752.
- Halloran ME, Auranen K, Baird S, Basta NE, Bellan SE, Brookmeyer R, et al. Simulations for designing and interpreting intervention trials in infectious diseases. BMC Med. 2017; 15(1):223. <u>https://doi.org/10. 1186/s12916-017-0985-3 PMID: 29287587</u>; PubMed Central PMCID: PMC5747936.
- Perez-Heydrich C, Hudgens MG, Halloran ME, Clemens JD, Ali M, Emch ME. Assessing effects of cholera vaccination in the presence of interference. Biometrics. 2014; 70(3):731–44. https://doi.org/10. 1111/biom.12184 PMID: 24845800; PubMed Central PMCID: PMC4239215.
- Morris TP, White IR, Crowther MJ. Using simulation studies to evaluate statistical methods. Stat Med. 2019; 38(11):2074–102. <u>https://doi.org/10.1002/sim.8086 PMID: 30652356</u>; PubMed Central PMCID: PMC6492164.
- Hitchings MD, Grais RF, Lipsitch M. Using simulation to aid trial design: Ring-vaccination trials. PLoS Negl Trop Dis. 2017; 11(3):e0005470. https://doi.org/10.1371/journal.pntd.0005470 PMID: 28328984; PubMed Central PMCID: PMC5378415.
- Camacho A, Eggo RM, Goeyvaerts N, Vandebosch A, Mogg R, Funk S, et al. Real-time dynamic modelling for the design of a cluster-randomized phase 3 Ebola vaccine trial in Sierra Leone. Vaccine. 2017; 35(4):544–51. https://doi.org/10.1016/j.vaccine.2016.12.019 PMID: 28024952.
- 10. Ontweka LN, Deng LO, Rauzier J, Debes AK, Tadesse F, Parker LA, et al. Cholera Rapid Test with Enrichment Step Has Diagnostic Performance Equivalent to Culture. PLoS One. 2016; 11(12): e0168257. https://doi.org/10.1371/journal.pone.0168257 PMID: 27992488; PubMed Central PMCID: PMC5167375.
- 11. Funk S. bpmodels: Analysing chain statistics using branching process models 2020. 0.1.0:[Available from: https://github.com/sbfnk/bpmodels.
- Hughes J. Running Power Simulations with the Power of R [Blog entry]. 2017 [updated Sept 27, 2017; cited 2021]. Available from: http://disjointedthinking.jeffhughes.ca/2017/09/power-simulations-r.
- Nelson EJ, Harris JB, Morris JG, Calderwood SB, Camilli A. Cholera transmission: the host, pathogen and bacteriophage dynamic. Nature reviews Microbiology. 2009; 7(10):693–702. https://doi.org/10. 1038/nrmicro2204 PMID: 19756008.
- Azman AS, Rumunu J, Abubakar A, West H, Ciglenecki I, Helderman T, et al. Population-level effect of cholera vaccine on displaced populations, South Sudan, 2014. Emerg Infect Dis. 2016; 22(6):1067–70. https://doi.org/10.3201/eid2206.151592 PMID: 27192187; PubMed Central PMCID: PMC4880069.
- Camacho A, Bouhenia M, Alyusfi R, Alkohlani A, Naji MAM, de Radigues X, et al. Cholera epidemic in Yemen, 2016–18: an analysis of surveillance data. Lancet Glob Health. 2018; 6(6):e680–e90. https:// doi.org/10.1016/S2214-109X(18)30230-4 PMID: 29731398; PubMed Central PMCID: PMC5952990.
- Emch M, Feldacker C, Islam MS, Ali M. Seasonality of cholera from 1974 to 2005: a review of global patterns. Int J Health Geogr. 2008; 7:31. Epub 2008/06/24. https://doi.org/10.1186/1476-072X-7-31 PMID: 18570659; PubMed Central PMCID: PMC2467415.
- Lloyd-Smith JO, Schreiber SJ, Kopp PE, Getz WM. Superspreading and the effect of individual variation on disease emergence. Nature. 2005; 438(7066):355–9. Epub 2005/11/18. <u>https://doi.org/10.1038/nature04153</u> PMID: 16292310; PubMed Central PMCID: PMC7094981.
- Ratnayake R, Finger F, Edmunds WJ, Checchi F. Early detection of cholera epidemics to support control in fragile states: estimation of delays and potential epidemic sizes. BMC Medicine. 2020; 18(1):397.

https://doi.org/10.1186/s12916-020-01865-7 PMID: 33317544; PubMed Central PMCID: PMC7737284.

- Parker LA, Rumunu J, Jamet C, Kenyi Y, Lino RL, Wamala JF, et al. Neighborhood-targeted and casetriggered use of a single dose of oral cholera vaccine in an urban setting: Feasibility and vaccine coverage. PLoS Negl Trop Dis. 2017; 11(6):e0005652. https://doi.org/10.1371/journal.pntd.0005652 PMID: 28594891; PubMed Central PMCID: PMC5478158.
- 20. Ouamba JP, Peyraud N, Mbarga NF, Ciglenecki I, Finger F, Ratnayake R, et al. Case-area targeted interventions for cholera control: experience from tail of cholera outbreak in Kribi, Cameroon. In: MSF UK, editor. MSF Scientific Days [Internet]; 2021 May 18–20; Médecins Sans Frontières UK, London, UK, 2021 [cited 2022 January 31]. Available from: https://msf.org.uk/sites/default/files/2021%20SD% 20Poster_CATI_CM144%20VF_ONJP.pdf.
- Fewtrell L, Kaufmann RB, Kay D, Enanoria W, Haller L, Colford JM Jr. Water, sanitation, and hygiene interventions to reduce diarrhoea in less developed countries: a systematic review and meta-analysis. Lancet Infect Dis. 2005; 5(1):42–52. https://doi.org/10.1016/S1473-3099(04)01253-8 PMID: 15620560.
- Roberts L, Chartier Y, Chartier O, Malenga G, Toole M, Rodka H, Keeping clean water clean in a Malawi refugee camp: a randomized intervention trial. Bulletin of the World Health Organization. 2001; 79 (4):280–7. PMID: 11357205.
- 23. Reveiz L, Chapman E, Ramon-Pardo P, Koehlmoos TP, Cuervo LG, Aldighieri S, et al. Chemoprophylaxis in contacts of patients with cholera: systematic review and meta-analysis. PloS one. 2011; 6(11): e27060. https://doi.org/10.1371/journal.pone.0027060 PMID: 22102873.
- Azman AS, Parker LA, Rumunu J, Tadesse F, Grandesso F, Deng LL, et al. Effectiveness of one dose of oral cholera vaccine in response to an outbreak: a case-cohort study. Lancet Glob Health. 2016; 4 (11):e856–e63. https://doi.org/10.1016/S2214-109X(16)30211-X PMID: 27765293.
- Hubbard AE, Ahern J, Fleischer NL, Van der Laan M, Lippman SA, Jewell N, et al. To GEE or not to GEE: comparing population average and mixed models for estimating the associations between neighborhood risk factors and health. Epidemiology. 2010; 21(4):467–74. https://doi.org/10.1097/EDE. 0b013e3181caeb90 PMID: 20220526.
- James G, Witten D, Hastie T, Tibshirani R. Generalized Additive Models: GAMs for Regression Problems. An Introduction to statistical learning: with applications in R. New York: Springer; 2013. p. 306–307.
- Roskosky M, Acharya B, Shakya G, Karki K, Sekine K, Bajracharya D, et al. Feasibility of a comprehensive targeted cholera intervention in the Kathmandu Valley, Nepal. Am J Trop Med Hyg. 2019; 100 (5):1088–97. https://doi.org/10.4269/ajtmh.18-0863 PMID: 30887946; PubMed Central PMCID: PMC6493959.
- Michel E, Gaudart J, Beaulieu S, Bulit G, Piarroux M, Boncy J, et al. Estimating effectiveness of casearea targeted response interventions against cholera in Haiti. Elife. 2019; 8. https://doi.org/10.7554/ eLife.50243 PMID: 31886768; PubMed Central PMCID: PMC7041943.